Author’s response to reviews

Title: Prevalence of attention-deficit/hyperactivity disorder symptoms and their associations with sleep schedules and sleep-related problems among preschoolers in Mainland China

Authors:

hui cao (caohui1230@126.com)
Shuang Yan (Yanqs12@126.com)
Chun Gu (Chunligu12@126.com)
Su Wang (Meilisu12@126.com)
Ling Ni (2635974413@qq.com)
Hui Tao (78964086@qq.com)
Ting Shao (1210702573@qq.com)
Ye Xu (xll0628@163.com)
Fang Tao (taofangbiao1@126.com)

Version: 4 Date: 19 Jan 2018

Author’s response to reviews:

Response to the comments

Dear Editor,

Re: “Prevalence of attention-deficit/hyperactivity disorder symptoms and their associations with sleep schedules and sleep-related problems among preschoolers in Mainland China” (BPED-D-16-00652 Version 4). Thank you very much for your letter of 18 January 2018, and the attached requirements. We have reviewed our manuscript carefully and revised the manuscript according to the editor comment.
1. In the Ethics and consent to participate section of the Declarations, we add this sentence “Informed consent obtained from all the participating caregivers of children was written”. (Page 366-367)

2. We add the “Declarations” heading in the revised manuscript. (Page 343)

Thanks again!

Sincerely yours  Hui Cao, Shuangqin Yan, Fangbiao Tao

19 January, 2018